Author’s response to reviews

Title: Suicide and No Axis I Psychopathology

Authors:

Carl Ernst (cp_ernst@yahoo.ca)
Aleksandra Lalovic (aleksandra.lalovic@mail.mcgill.ca)
Alain Lesage (alesage@ssss.gouv.qc.ca)
Monique Seguin (monique.seguin@ugo.ca)
Michel Tousignant (tousignant.michel@uqam.ca)
Dr Gustavo Turecki (gustavo.turecki@mcgill.ca)

Version: 2 Date: 5 Mar 2004

Editor
BMC Psychiatry

March 5, 2004

Dear Editor,

Re: MS: 1808031541278149
Suicide and No Axis I Psychopathology

Please find attached the revised version of the above-referred manuscript. The only comments raised by the reviewers were as follows:

To understand the findings, the authors could have discussed whether excessive gambling is found in the group with Axis I disorders. In addition, the discussion should include some review of other research that has found a relationship between excessive gambling and suicide.

We have revised the discussion of the manuscript to address the reviewer’s comments. The second paragraph of the revised discussion refers to links reported in the literature between pathological gambling and suicidal behavior and discusses why it was not possible to directly compare levels of excessive gambling between the NA1S and A1S groups.

Should you require additional information or further modifications, please do not hesitate to contact me.

Sincerely,

Gustavo Turecki MD PhD
Director,
McGill Group for Suicide Studies
Douglas Hospital
McGill University
6875 LaSalle blvd
Montreal QC H4H 1R3
Canada
Phone: (514) 761-6131 xt.2369
Fax: (514) 762-3011
gustavo.turecki@mcgill.ca